Reviewer #1: The final paragraph of the text should be revised once again by a language editing service.

Answer: I have revised the final paragraph of my text by a language editing service. And the language certificate will be uploaded also.

Reviewer #2: Signed Consent Document is not in English. CARE Checklist–2016 should be scanned in order to be easily read.

Answer: Patients informed consent forms in China are generally signed in Chinese, and the guidelines for author clearly state that informed consent forms in the native language of each country should be submitted. If the English version is really needed, it may take a few more days for the document to be fair. I will upload a new and clear CARE Checklist–2016.